Author’s response to reviews

Title: Role of nurses in medication management at the end of life: a qualitative interview study

Authors:

Bregje Huisman (b.huisman@amsterdamumc.nl)
Eric Geijteman (e.geijteman@erasmusmc.nl)
Marianne Dees (Marianne.Dees@radboudumc.nl)
Noralie Schonewille (noralieongschonewille@gmail.com)
Margriet Wieles (consultatie@kuria.nl)
Lia van Zuylen (c.vanzuylen@erasmusmc.nl)
Karolina Szadek (km.szadek@amsterdamumc.nl)
Agnes van der Heide (a.vanderheide@erasmusmc.nl)

Version: 3 Date: 01 May 2020

Author’s response to reviews:

1. We note that table 1 contains potentially identifiable information. It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper (http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9).

Currently, the age and gender information in conjunction with other details, may compromise patient anonymity.

Please amend table 1 to address this issue by removing this information or by providing ages as age-ranges, and removing the age (or sex) information.

Reply: We adjusted table 1, with the age in a range.

2. Please remove the COREQ checklist and any references to this from the manuscript, as at this stage the COREQ checklist is no longer necessary.

Reply: We adjusted this in the tekst, the references and removed the supplement.
3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Reply: done.